\begin{thebibliography}{10}

\bibitem{Agarwal90a}
A.~Agarwal, B.-H. Lim, D.~Kranz, and J.~Kubiatowicz.
\newblock {APRIL: A Processor Architecture for Multiprocessing}.
\newblock In {\em Proc. 17th Ann. Intl. Symp. on Computer Architecture,
  Seattle, WA}, pages 104--114, May 1990.

\bibitem{Almasi89}
G.~S. Almasi and A.~Gottlieb.
\newblock {\em Highly Parallel Computing}.
\newblock Benjamin/Cummings, Redwood City, CA, 1989.

\bibitem{Alverson90a}
R.~Alverson, D.~Callahan, D.~Cummings, B.~Koblenz, A.~Porterfield, and
  B.~Smith.
\newblock {The Tera Computer System}.
\newblock In {\em Proc. Intl. Conf. on Supercomputing, Amsterdam}, June 1990.

\bibitem{Arvind86a}
Arvind and D.~E. Culler.
\newblock {Dataflow Architectures}.
\newblock In {\em {Annual Reviews in Computer Science}}, volume~1, pages
  225--253. Annual Reviews Inc., Palo Alto, CA, USA, 1986.

\bibitem{Arvind83a}
Arvind, D.~E. Culler, R.~A. Iannucci, V.~Kathail, K.~Pingali, and R.~E. Thomas.
\newblock {The Tagged Token Dataflow Architecture}.
\newblock Technical report, MIT Lab. for Computer Science, Aug. 1983.
\newblock Revised October, 1984.

\bibitem{Arvind87c}
Arvind and R.~A. Iannucci.
\newblock {Two Fundamental Issues in Multiprocessing}.
\newblock In {\em Proc. DFVLR - 1987 Conf. on Parallel Processing in Science
  and Engineering, Bonn-Bad Godesberg, W. Germany (Springer-Verlag LNCS 295)},
  June 1987.

\bibitem{Arvind90a}
Arvind and R.~S. Nikhil.
\newblock {Executing a Program on the MIT Tagged-Token Dataflow Architecture}.
\newblock {\em IEEE Trans. on Computers}, 39(3):300--318, Mar. 1990.

\bibitem{Brooks89a}
E.~Brooks.
\newblock {The Attack of the Killer Micros}, 1989.
\newblock Teraflop Computing Panel, Supercomputing '89, Reno, Nevada.

\bibitem{Buehrer87a}
R.~Buehrer and K.~Ekanadham.
\newblock {Incorporating Dataflow Ideas into von Neumann Processors for
  Parallel Execution}.
\newblock {\em IEEE Trans. on Computers}, C-36(12):1515--1522, Dec. 1987.

\bibitem{Culler91a}
D.~E. Culler and G.~M. Papadopoulos.
\newblock {The Explicit Token Store}.
\newblock {\em J. Parallel and Distributed Computing}, Dec. 1990.

\bibitem{Culler90a}
D.~E. Culler, A.~Sah, K.~E. Schauser, T.~{von Eicken}, and J.~Wawrzynek.
\newblock {Fine-grain Parallelism with Minimal Hardware Support: A
  Compiler-Controlled Threaded Abstract Machine}.
\newblock In {\em Proc. 4th Intl. Conf. on Arch. Support for Programming
  Languages and Systems (ASPLOS), Santa Clara, CA}, pages 164--175, Apr. 1991.

\bibitem{Dally87a}
W.~Dally, L.~Chao, A.~Chien, S.~Hassoun, W.~Horwat, J.~Kaplan, P.~Song,
  B.~Totty, and S.~Wills.
\newblock {Architecture of a Message-Driven Processor}.
\newblock In {\em Proc. 14th. Ann. Intl. Symp. on Computer Architecture,
  Pittsburgh, PA}, pages 189--196, June 1987.

\bibitem{Dennis91a}
J.~B. Dennis.
\newblock {The Evolution of ``Static'' Data-Flow Architecture}.
\newblock In {\em Advanced Topics in Data-flow Computing, J-L. Gaudiot and L.
  Bic (eds.)}, pages 35--91. Prentice-Hall, 1991.

\bibitem{Dennis88a}
J.~B. Dennis and G.~R. Gao.
\newblock {An Efficient Pipelined Dataflow Processor Architecture}.
\newblock In {\em Proc. Supercomputing, Orlando, FL}, pages 368--373, Nov.
  1988.

\bibitem{Grafe88a}
V.~Grafe, G.~Davidson, J.~Hoch, and V.~Holmes.
\newblock {The Epsilon Dataflow Processor}.
\newblock In {\em Proc. 16th. Ann. Intl. Symp. on Computer Architecture,
  Jerusalem, Israel}, pages 36--45, May 1989.

\bibitem{Gurd85}
J.~R. Gurd, C.~Kirkham, and I.~Watson.
\newblock {The Manchester Prototype Dataflow Computer}.
\newblock {\em Comm. of the ACM}, 28(1):34--52, Jan. 1985.

\bibitem{Halstead88a}
R.~H. {Halstead Jr.} and T.~Fujita.
\newblock {MASA: A Multithreaded Processor Architecture for Parallel Symbolic
  Computing}.
\newblock In {\em Proc. 15th Ann. Intl. Symp. on Computer Architecture,
  Honolulu, Hawaii}, June 1988.

\bibitem{Henry91a}
D.~S. Henry and C.~F. Joerg.
\newblock {The Network Interface Chip}.
\newblock Technical Report CSG Memo 331, MIT Laboratory for Computer Science,
  545 Technology Square, Cambridge MA 02139, June 1991.

\bibitem{Hiraki87b}
K.~Hiraki, K.~Nishida, S.~Sekiguchi, T.~Shimada, and T.~Yuba.
\newblock {The SIGMA-1 Dataflow Supercomputer: A Challenge for New Generation
  Supercomputing Systems}.
\newblock {\em J. of Information Processing}, 10(4):219--226, 1987.

\bibitem{Iannucci88a}
R.~A. Iannucci.
\newblock {Toward a Dataflow/von Neumann Hybrid Architecture}.
\newblock In {\em Proc. 15th Ann. Intl. Symp. on Computer Architecture,
  Honolulu, Hawaii}, pages 131--140, June 1988.

\bibitem{Lenoski90a}
D.~Lenoski, J.~Laudon, K.~Gharachorloo, A.~Gupta, and J.~Hennessy.
\newblock {The Directory-Based Cache Coherence Protocol for the DASH
  Multiprocessor}.
\newblock In {\em Proc. 17th. Ann. Intl. Symp. on Computer Architetecture,
  Seattle, WA}, pages 148--159, May 1990.

\bibitem{Maurer88}
P.~M. Maurer.
\newblock {Mapping the Data Flow Model of Computation onto an Enhanced von
  Neumann Processor}.
\newblock In {\em Proc. Intl. Conf. on Parallel Processing}, pages 235--239,
  Aug. 1988.

\bibitem{Nikhil90e}
R.~S. Nikhil.
\newblock {The Parallel Programming Language Id and its Compilation for
  Parallel Machines}.
\newblock In {\em Proc. Wkshp. on Massive Parallelism, Amalfi, Italy}, October
  1989.

\bibitem{Nikhil88d}
R.~S. Nikhil and Arvind.
\newblock {Can dataflow subsume von Neumann computing?}
\newblock In {\em Proc. 16th. Ann. Intl. Symp. on Computer Architecture,
  Jerusalem, Israel}, pages 262--272, May 1989.

\bibitem{Papadopoulos88a}
G.~M. Papadopoulos.
\newblock {\em {Implementation of a General Purpose Dataflow Multiprocessor}}.
\newblock MIT Press, 1991.

\bibitem{Papadopoulos91a}
G.~M. Papadopoulos and K.~R. Traub.
\newblock {Multithreading: A Revisionist View of Dataflow Architectures}.
\newblock In {\em Proc. 18th. Intl. Symp. on Computer Architecture, Toronto},
  Mar. 1991.

\bibitem{Pfister85a}
G.~Pfister, W.~Brantley, D.~George, S.~Harvey, W.~Kleinfelder, K.~McAuliffe,
  E.~Melton, V.~Norton, and J.~Weiss.
\newblock {The IBM Research Parallel Processor Prototype (RP3): Introduction
  and Architecture}.
\newblock In {\em Proc. Intl. Conf. on Parallel Processing}, pages 764--771,
  Aug. 1985.

\bibitem{Sakai89a}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, and T.~Yuba.
\newblock {An Architecture of a Dataflow Single Chip Processor}.
\newblock In {\em Proc. 16th Ann. Intl. Symp. on Computer Architecture,
  Jerusalem, Israel}, pages 46--53, May 1989.

\bibitem{Schauser91}
K.~E. Schauser, D.~E. Culler, and T.~{von Eicken}.
\newblock {Compiler-Controlled Multithreading for Lenient Parallel Languages}.
\newblock In {\em Proc. 5th ACM Conf. on Functional Programming Languages and
  Computer Architecture, Cambridge, MA (Springer-Verlag LNCS 523)}, pages
  50--72, Aug. 1991.

\bibitem{Smith78}
B.~J. Smith.
\newblock {A Pipelined, Shared Resource MIMD Computer}.
\newblock In {\em Proc. Intl. Conf. on Parallel Processing}, pages 6--8, 1978.

\end{thebibliography}

% Introduction section for ISCA paper

\section{Introduction}

There appears to be widespread consensus that general purpose
supercomputers of the future will be Massively Parallel Architectures
(MPA's) consisting of a number of nodes in a high speed, regular
interconnection network.  This view has been eloquently summarized in
the slogan ``attack of the killer micros'' by Eugene Brooks of
Lawrence Livermore Laboratories \cite{Brooks89a}.  In this paper, we
focus on the organization of each node in such an MPA.

In an MPA, inter-node communication latency is expected to be high and
caching global locations is difficult.  In Section \ref{sec-analysis}
we present this issue in concrete terms by describing two specific
latency-related problems that must be performed well by MPA nodes.  In
Section \ref{sec-solutions} we systematically develop the logical node
organization of {\archname}, that addresses these problems.
Pronounced {\em Start}, it is named after the principal communication
instruction in the machine.  It can also be read as an acronym for
``multi- (*) Threaded'', a principal feature of the machine.  In
Section \ref{sec-abstract-model} we present some instruction set
details of the logical organization of {\archname}, and in Section
\ref{sec-daxpy} we discuss the coding of DAXPY.
In Section \ref{sec-implementation} we discuss possibilities for
physical implementation of {\archname}, and we conclude in Section
\ref{sec-compiling} with some discussion of compilation issues.

{\archname} is the latest step in the evolution of dynamic dataflow
architectures, starting with the MIT Tagged-Token Dataflow
Architecture (TTDA,~\cite{Arvind83a,Arvind90a}), the Manchester
dataflow machine~\cite{Gurd85} and the ETL SIGMA-1~\cite{Hiraki87b}.
In 1987-88, the associative waiting-matching store of these machines
was replaced by an explicitly managed, directly addressed token store,
by Papadopoulos and Culler in the Monsoon
architecture~\cite{Papadopoulos88a,Culler91a}.  Nikhil and Arvind's
P-RISC architecture~\cite{Nikhil88d} then split the ``complex''
dataflow instructions into separate synchronization, arithmetic and
fork/control instructions, eliminating the necessity of presence bits
on the token store (also known as ``frame memory'').  P-RISC also
permitted the compiler to assemble instructions into longer threads,
replacing some of the dataflow synchronization with conventional
program counter-based synchronization (in retrospect, Monsoon could
also be viewed in this light~\cite{Papadopoulos91a}).  {\archname} is
the latest step in this direction, eliminating the prior distinction
between processor and I-structure nodes, and attempting also to be
compatible with conventional MPA's based on von Neumann processors.

Throughout this evolution, our {\em system view\/} of an MPA (\ie\
viewing each node of an MPA as a black box) has remained constant, and
is in fact quite unique to dataflow architectures.  Inter-node traffic
consists of tokens (\ie\ messages) which contain a {\em tag\/} (or
{\em continuation}, a pair comprising a context and instruction
pointer) and one or more values.  All inter-node communications are
performed with {\em split-phase\/} transactions (request and response
messages); processors never block when issuing a remote request, and
the network interface for message handling is well integrated into the
processor pipeline.  Thus, TTDA, Monsoon, P-RISC and {\archname} are
properly viewed as steps in the evolution of the {\em node
architecture\/} of dataflow MPA's.

Important architectural influences in our thinking have been the
seminal Denelcor HEP~\cite{Smith78} and Iannucci's Dataflow/von
Neumann Hybrid architecture~\cite{Iannucci88a}, which incorporated
some dataflow ideas into von Neumann processors for MPA's (similar
ideas were also expressed by Buehrer and Ekanadham
~\cite{Buehrer87a}).  ETL's EM-4~\cite{Sakai89a}, Sandia's
Epsilon~\cite{Grafe88a}, Dennis and Gao's Argument-Fetching Dataflow
machine~\cite{Dennis88a}, and Dennis' multithreaded
architecture~\cite{Dennis91a} are also dataflow machines supporting
threads.  Finally, the Network Interface Chip by Henry and
Joerg~\cite{Henry91a} has had a direct influence on our implementation
of {\archname}.

Software considerations, \ie\ compiling, have played an equally
important role in the design of {\archname}.  Compilation of the
high-level parallel programming language Id has been central to our
research since the beginning~\cite{Arvind90a}.  Nikhil,
in~\cite{Nikhil90e}, and Culler {\em et al\/} in~\cite{Culler90a}
proposed P-RISC-like compilation models extended to allow additional
instruction sequencing constraints, making explicit the notion of a
thread.  Culler's model, the Threaded Abstract Machine (TAM) has been
implemented by his group and has been in use for almost two years,
providing a wealth of data on the nature of threads, locality and
synchronization~\cite{Schauser91}.  This experience has directly
influenced the {\archname} design.

Many researchers have arrived at similar mechanisms starting from
other vantage points (\eg\ Halstead and Fujita's
MASA~\cite{Halstead88a} and Maurer's SAM \cite{Maurer88}).  There are
very interesting comparisons to be made between {\archname} and recent
multithreaded architectures, notably the Tera Computer
System~\cite{Alverson90a} (a descendant of the HEP),
Alewife~\cite{Agarwal90a} and the J-Machine with Message Driven
Processors (MDPs)~\cite{Dally87a}.  Undoubtedly, aspects of these
machines will influence concrete {\archname} designs.
